Author’s response to reviews

Title: Maternal caffeine intake during pregnancy is associated with birth weight but not with gestational length: results from a large prospective observational cohort study

Authors:

Verena Sengpiel (verena.sengpiel@obgyn.gu.se)
Elisabeth Elind (elielind@online.no)
Jonas Bacelis (jonas.bacelis@gmail.com)
Staffan Nilsson (staffan@chalmers.se)
Jakob Grove (grove@hum-gen.au.dk)
Ronny Myhre (ronny.myhre@fhi.no)
Margaretha Haugen (margaretha.haugen@fhi.no)
Helle M Meltzer (helle.margrete.meltzer@fhi.no)
Jan Alexander (jan.alexander@fhi.no)
Bo Jacobsson (bo.jacobsson@obgyn.gu.se)
Anne-Lise Brantsæter (anne.lise.brantsaeter@fhi.no)

Version: 3 Date: 6 November 2012

Author’s response to reviews: see over
Dear Dr Lin Lee

Thank you for accepting our paper “Maternal caffeine intake during pregnancy is associated with birth weight but not with gestational length: results from a large prospective observational cohort study” (MS: 4584083108059329) for publication in BMC Medicine.

You will find a revised version of the manuscript attached. Tables have been formatted as requested; all other changes in the authors’ contribution section and reference section have been highlighted in blue.

Yours sincerely,

Verena Sengpiel, on behalf of all authors

Verena Sengpiel, MD, PhD
Dept of Obstetrics and Gynaecology
Sahlgrenska University Hospital/Östra
SE-416 85 Gothenburg
Sweden